Author’s response to reviews

Title: Reimbursement for reconstruction by tissue transfer - A European comparison

Authors:

Oliver Lotter (oliver.lotter@freenet.de)  
William A Townley (wtownley@hotmail.com)  
Hans-Eberhard Schaller (Schaller@bgu-tubingen.de)  
Patrick Jaminet (patrickjaminet@gmx.de)

Version: 2 Date: 26 August 2012

Author’s response to reviews:

Dear Sirs,

please find the Ethics statement in the Methods section of this paper in red letters.

Yours sincerely

Oliver Lotter